**Author's response to reviews**

**Title:** Impact of Age on Outcome after Colorectal Cancer Surgery in the Elderly - a developing country perspective

**Authors:**

Muhammad Rizwan Khan (doctormrkhan@yahoo.com)
Hassaan Bari (hassaan_bari@yahoo.com)
Syed Nabeel Zafar (nabeel.zafar@aku.edu)
Syed Ahsan Raza (s.ahsanraza@gmail.com)

**Version:** 3  **Date:** 14 August 2011

**Author's response to reviews:**

The following changes have been made in the manuscript, as requested by the editorial team:

1. Major change: the pattern of tables has been changed, as advised.
2. Minor changes:
   i. Author’s titles have been removed.
   II. Additional files have been removed.